

Initiated by Deutsche Post Foundation

DISCUSSION PAPER SERIES

IZA DP No. 16687

Automatability of Occupations, Workers' Labor-Market Expectations, and Willingness to Train

Philipp Lergetporer Katharina Wedel Katharina Werner

DECEMBER 2023



Initiated by Deutsche Post Foundation

DISCUSSION PAPER SERIES

IZA DP No. 16687

Automatability of Occupations, Workers' Labor-Market Expectations, and Willingness to Train

Philipp Lergetporer Technical University Munich and IZA

Katharina Wedel ifo Institute

Katharina Werner ifo Institute

DECEMBER 2023

Any opinions expressed in this paper are those of the author(s) and not those of IZA. Research published in this series may include views on policy, but IZA takes no institutional policy positions. The IZA research network is committed to the IZA Guiding Principles of Research Integrity.

The IZA Institute of Labor Economics is an independent economic research institute that conducts research in labor economics and offers evidence-based policy advice on labor market issues. Supported by the Deutsche Post Foundation, IZA runs the world's largest network of economists, whose research aims to provide answers to the global labor market challenges of our time. Our key objective is to build bridges between academic research, policymakers and society.

IZA Discussion Papers often represent preliminary work and are circulated to encourage discussion. Citation of such a paper should account for its provisional character. A revised version may be available directly from the author.

ISSN: 2365-9793

IZA – Institute of Labor Economics

Schaumburg-Lippe-Straße 5–9	Phone: +49-228-3894-0	
53113 Bonn, Germany	Email: publications@iza.org	www.iza.org

ABSTRACT

Automatability of Occupations, Workers' Labor-Market Expectations, and Willingness to Train^{*}

We study how beliefs about the automatability of workers' occupation affect labor-market expectations and willingness to participate in further training. In our representative online survey, respondents on average underestimate the automation risk of their occupation, especially those in high-automatability occupations. Randomized information about their occupations' automatability increases respondents' concerns about their professional future, and expectations about future changes in their work environment. The information also increases willingness to participate in further training, especially among respondents in highly automatable occupation (+five percentage points). This uptick substantially narrows the gap in willingness to train between those in high- and low-automatability occupations.

JEL Classification:	J24, O33, I29, D83
Keywords:	automation, further training, labor-market expectations, survey experiment, information

Corresponding author:

Philipp Lergetporer Technical University Munich Wedel Germany E-mail: philipp.lergetporer@tum.de

^{*} For helpful comments, we would like to thank David Autor, Peter Blair, Adrian Lerche, Armando Miano, Stephen L. Ross, Giuseppe Sorrenti, Simon Wiederhold, Ludger Woessmann, and seminar participants at the 14th Interdisciplinary GradAB workshop in Nürnberg, the CRC Rationality and Competition Meeting, the German Economic Association, and its economics of education group, the ifo Center for Economics of Education, the CESifo / ifo Junior Workshop on the Economics of Education, the CESifo Venice Summer Institute, EALE conference, and the TUM Campus Heilbronn Global Technology Forum. We are most grateful to Vera Freundl and Franziska Kugler for their help in preparing the survey and to Elias Farnleitner, Silas Kaatz, Jakob Sträter and Simon Zürneck for excellence research assistance. Financial support by the Leibniz Competition (SAW-2014-ifo-2) and the German Science Foundation (CRC TRR 190) is gratefully acknowledged. The experiment presented in this study is registered in the AEA RCT Registry (AEARCTR-0009464).

1. Introduction

Technological progress is a key driver of economic growth (Acemoglu 2009; Jones 2002; Aghion and Howitt 1992). Changes in technology often bring about drastic changes in the demand for different input factors. One notable example is the ongoing digital transformation: technologies like artificial intelligence have already transformed the skill demand in many professions, rendering existing skills and sometimes entire professions obsolete (OECD 2021; Dachs 2018; OECD 1998). At the same time, there is an increasing demand for occupations and skills that complement new technologies.¹ Because of these new and changing technological opportunities and the rising automatability of occupations, the knowledge that workers have acquired becomes outdated at an ever-faster rate. Therefore, further training throughout one's working life is crucial for workers to keep pace with, and benefit from, structural change on the labor market (e.g., Innocenti and Golin 2022; Bessen 2019).

In this paper, we study whether workers are aware of their occupations' automatability, and how factual information about occupations' automatability affects labor-market expectations and willingness to participate in further training. Workers in occupations with high automatability are strongly underrepresented in further training initiatives in many countries (Heß et al. 2019; OECD 2021), whereas their employment prospects would benefit most from further training.² The inequality in further-training participation between individuals in high-and low-automatability occupations is not yet fully understood. Based on the observation that people are often misinformed about labor-market relevant facts, like their probability to find a new job, wages, or outside options (Jäger et al. 2021; Mueller et al. 2021), we hypothesize that workers' misperceptions about their occupations' automatability may contribute to low training participation of workers in high-automatability occupations.³ If workers underestimate their occupations' automatability, they may underinvest in keeping their skills up-to-date,

¹ For Germany, the country we study, the Federal Ministry of Labour and Social Affairs estimates that 5.3 million jobs will be lost by 2040, while 3.6 million new jobs will be created (Bundesministerium für Arbeit und Soziales, 2021).

² Several studies show that workers in high-automatability occupations have worse employment outcomes and lower wage growth, which highlights the importance of studying their obstacles to participating in further training (Georgieff and Milanez 2021; Montobbio et al. 2022; Acemoglu and Restrepo 2020b; Schmidpeter and Winter-Ebmer 2021; Dauth et al. 2021).

³ Complementary reasons for non-participation in further training include, e.g., stigmatization (since further training possibilities are sometimes offered by the Federal Employment Agency, which most people in Germany associate with unemployment), failure to recognize potential benefits, high costs, or lack of time resources (e.g., van den Berg et al. 2019; Müller and Wenzelmann 2020; Osiander and Stephan 2020). We report further details on barriers to participation in section 4.3.

potentially jeopardizing their labor-market success amidst ongoing labor-market transformations.

Previous literature has mostly focused on the effects of automation on labor-market outcomes. In contrast, little is known about how workers perceive their occupations' automatability, and how these beliefs affect labor-market expectations and willingness to train. We address this gap by (i) documenting beliefs about automatability in an online sample of the German population and (ii) studying experimentally how providing information about their occupations' automatability affects workers' beliefs, labor-market expectations, and attitudes towards further training. By examining the effects of randomized information provision, we can isolate the causal impact of correcting potentially biased beliefs on workers' expectations and willingness to training.

We implemented our experiment in a large online survey (N=3,012) to represent the German adult population. In the survey, we first elicited respondents' prior beliefs about the automatability of their occupations. We then provided the randomly selected treatment group of respondents with personalized information about their occupations' automatability. The remaining respondents served as the uninformed control group. We drew the information about occupations' automatability from the "IAB-Job-Futuromat" (https://job-futuromat.iab.de/) of the Research Institute of the Federal Employment Agency (IAB). The IAB-Job-Futuromat calculates each occupation's automatability by determining the share of its core tasks that could potentially be performed fully automatically, either by a computer or computer-controlled machine, according to expert evaluations (Dengler and Matthes 2018a). Finally, we measured (i) respondents' labor-market expectations about their professional future, (ii) their likelihood to participate in further training (e.g., skill enhancement programs) and retraining (i.e., career transition courses) and (iii) their willingness to accept a reduced wage during their further training period. By comparing responses between the uninformed control and the informed treatment group, we evaluate how factual information about the automatability of workers' occupations affects these outcomes.

We find that, on average, respondents underestimate the automatability of their own occupation. Crucially, the misperception is particularly large for those in occupations with high automatability. Descriptively, within the control group, respondents who believe that their occupations' automatability is low state the lowest likelihood of participating in further training. Furthermore, there is a positive correlation between respondents' beliefs about their occupations' automatability and their likelihood to participate in retraining. The information

experiment allows us to investigate how exogenously shifting beliefs about automatability causally affects our outcomes of interest.

Experimental results show that information provision about the automatability of one's occupation affects labor-market expectations. Respondents become 9.7 percent of a standard deviation more concerned about their future work and are more likely to expect changes in their work environment (+13.0 percent of a standard deviation). Furthermore, the information treatment increases the stated likelihood of participating in further training and retraining by 5.8 and 12.5 percent respectively, relative to the mean. It also increases the fraction of wages that respondents are willing to forgo to participate in further training.

In line with our findings on misperceptions, the treatment effects for respondents in occupations with high automatability are larger than those with low automatability. Controlgroup respondents with high automatability express a 37.6 percent likelihood that they will participate in further training. Treated respondents in these occupations state a 4.6 percentage points (5.2 percentage points) higher likelihood to participate in further training (retraining). Conversely, for those in low-automatability occupations, the treatment effects on these outcomes are minimal and not statistically significant at -0.1 and 1.5 percentage points. Taken together, information provision reduces the control-group gap in the willingness to participate in further training between those in high- and low-automatability occupations by 95.5 percent, and fully closes the gap in willingness to participate in retraining. The same pattern emerges when considering respondents' willingness to accept a reduced wage to participate in further training. These findings suggest that misperceptions regarding occupations' automatability significantly contribute to the observed low training participation rates among workers in occupations most susceptible to technological change and automation.

Further results show that reluctance to participate in further training is not due to ignorance of the benefits. A large majority (76.7 percent) of respondents agree that further training is useful for keeping pace with structural change. Similarly, 66.2 percent agree that the future need for further training will increase for all employees, and 62.5 percent agree that everyone affected by structural change should participate in further training. Respondents state that the main reasons for not participating in further training are financial constraints (45.4 percent), lack of employer support (45.0 percent), and time constraints (35.2 percent). Answers on these outcomes are unaffected by treatment-group status. Treated respondents also request additional information about further training programs and finance options at the same rate as respondents in the control group.

The remainder of this paper is structured as follows. Section 2 provides an overview of the existing literature. Section 3 presents the institutional background, the study setup, and the empirical strategy. Section 4 presents descriptive and experimental results. Section 5 concludes.

2. Related Literature

Our study contributes to several strands of the existing literature. First, we relate to the literature on the effects of technology and automation on labor-market outcomes. Existing studies in this field have predominantly used non-survey data to examine the impact of automation on employment and wages. Acemoglu and Restrepo (2018, 2020a) highlight two key effects of new technologies on jobs and wages: the displacement effect, where robots replace human tasks, and the productivity effect, boosting productivity and thereby increasing labor demand in non-automated roles, leading to new jobs and tasks (Acemoglu and Restrepo 2019).⁴ Several studies have empirically examined the effect of (industrial) robots on employment outcomes (Acemoglu and Restrepo 2020a; Graetz and Michaels 2018; Autor and Salomons 2018; Dauth et al. 2021). Most studies find no effect on total employment and a positive impact on productivity and wages (Graetz and Michaels 2018; Dauth et al. 2021).⁵ The overall zero employment effect can mask substantial displacement and re-allocation effects (Arntz et al. 2020; Dauth et al. 2021)). Considering these findings, some researchers suggest automation may cause more worker transitions than mass unemployment (Bessen et al. 2020). Nonetheless, whether transitioning workplaces or facing unemployment, workers need to retrain, often learning new skills and switching occupations or industries. Participation in further-training programs is therefore vital for ensuring matching of labor demand and supply in rapidly changing labor markets. We contribute to this literature by using survey data and focusing specifically on the link between workers' beliefs about automatability and their labormarket expectations, as well as their willingness to (re)train.

Second, our paper adds to the literature that examines how workers (with varying skill levels) respond differently to technological change. For example, Blanas et al. (2019) argue

⁴ Recent studies on the emergence of "new work" and the task-based approach discuss how automation and computerization reallocate many human tasks to machine tasks, expanding the set of tasks performed by capital (Acemoglu and Restrepo 2022), and how they complement educated workers (Autor 2022). The introduction of new job tasks or job categories requires specialized human expertise, more education and workforce training (Autor et al. 2022; Autor 2022).

⁵ An exception is the paper by Acemoglu and Restrepo (2020a), which finds negative effects of robots on employment and wages across U.S. commuting zones.

that some workers respond to automation by transitioning to low-paid occupations where tasks are difficult to replace with machines, while other workers acquire new skills that complement machines and allow them to work in high-paid occupations. Cortes (2016) examines the effects of routine-biased technical change⁶ on workers' occupational transitions and wages, finding that especially low-ability routine workers tend to switch to non-routine manual tasks, while high-ability routine workers switch to non-routine cognitive occupations. We add to this literature by investigating how workers' misperceptions about the automatability of their occupations – a factor not previously studied – contributes to lower skill investments, especially among workers in high-automatability occupations. Other studies have examined barriers to training participation, such as misjudgment of potential benefits, high costs, or lack of time resources (e.g., van den Berg et al. 2019; Müller and Wenzelmann 2020; Osiander and Stephan 2020). We complement this by explicitly focusing on differences in participation rates across subgroups of workers and examining the influence of workers' beliefs about automatability on their willingness to train.

Moreover, our study relates to the literature on the relationship between automation and further training. Heß et al. (2019) find that further-training participation strongly varies with the level of routine tasks in employees' occupations: workers with the most routine tasks participate less (27 percent) compared to those with fewer routine tasks (41 percent). Innocenti and Golin (2022) show that 30 percent of respondents in their international sample are worried about being replaced by machines or algorithms and that workers' fear of automation is positively associated with their intentions to invest in training activities. Nedelkoska and Quintini (2018) use data from the Programme for the International Assessment of Adult Competencies (PIAAC) and find that workers in occupations at risk of automation have lower on-the-job and outside training participation rates. While these studies are primarily descriptive, we add to the literature by experimentally examining the causal link between automatability perceptions and further-training participation.

Methodologically, our paper relates to the literature leveraging survey experiments to examine how information affects public preferences (e.g., Cruces et al. 2013; Kuziemko et al. 2015; Bursztyn 2016). The experimental literature on information provision about automation and the future of work is closest to our study. Jeffrey (2021) shows that priming respondents

⁶ Since the 1980s, technological changes have led to the occurrence of more machines and computers that mainly perform routine tasks. Therefore, mainly routine workers are substituted. This hypothesis is termed routine-biased technological change (Cortes 2016). Earlier literature hypothesized that technological change is skill-biased, which favored high-skilled workers without distinguishing between tasks and skills (e.g., Katz and Murphy 1992; Autor et al. 1998).

with the notion that automation could lead to unfair disruptions in the labor market can increase preferences for redistribution. Arntz et al. (2022) provide two pieces of aggregate information about labor-market effects of automation (i.e., no aggregate employment losses, employment shift from unskilled to skilled workers). They find that information about zero net employment effects reduces concerns about automation but does not affect stated labor-market behavior and donations to NGOs. In comparison to these studies, our contribution lies in providing respondents with personalized, occupation-specific information about the automatability of their occupation. Closest to our paper, Golin and Rauh (2022) study how informing respondents about their job-loss probability affects policy preferences and different labor-market intentions. We extend this evidence by (i) focusing on automatability more generally and (ii) providing an in-depth analysis of how this type of information affects different dimensions of workers' labor-market beliefs and training intentions. This is particularly relevant for many contexts where automatability leads to changes on the labor market that extend beyond unemployment.

3. Background, Study Setup, and Empirical Strategy

In this section, we present the institutional background of further training in Germany (section 3.1), our data source, the ifo Education Survey (section 3.2), the experiment (section 3.3), the econometric model (section 3.4), and the sample description (section 3.5).

3.1. Institutional Background

In Germany, the majority of all further-training activities (72 percent) take place within the company (BMBF 2018). The average duration of in-company further training is 29 hours per training, which is shorter than the average individual job-related training outside the company (153 hours per training) (BMBF 2018). Based on the ifo Education Survey 2022, which serves as our data source (see section 3.2), 63 percent of respondents reported past participation in further training, revealing that over one third (37 percent) had not engaged in any additional training.

Overall, there are approximately 18,000 public and private further training providers (Bundesinstitut für Berufsbildung 2020). The training sector is governed by a variety of regulatory levels and legal foundations, including collective bargaining agreements, company agreements, laws, and state-level regulations. The responsibility and financial burden for further training are shared among companies, workers, and the public sector. The 'Work-of-Tomorrow Law' (Arbeit-von-morgen-Gesetz), a new German regulation introduced in 2020,

stipulates that further training exceeding 120 hours may receive subsidies from the Federal Employment Agency. Depending on the company's size, the Agency can finance up to 100 percent of the training costs. This law is specifically aimed at people with occupations that can be replaced by technology or are otherwise affected by structural change. For more information about the institutional background, see Appendix A.

3.2. Data

We use data from the ifo Education Survey 2022, a large-scale annual opinion survey on education policy in Germany (Freundl et al. 2022). The survey company Talk Online conducted sampling and polling of the online survey in May/June 2022. Overall, the survey encompassed questions related to education policy, focusing on topics related to structural change and lifelong learning.⁷ In addition, we elicited sociodemographic background characteristics at the end of the survey. Median completion time is 16 minutes, and the item non-response rate is low, with a maximum of 2.9 percent for the questions used in this study. We restrict our sample to respondents who are currently employed. Since the sample was drawn to represent the German population using quotas for gender, age, state, education level, and employment status, our data cover a broad sample of the German working population from different occupational fields with different requirement levels (see section 3.4). The overall sample size is 3,012 respondents.

3.3. The Experiment

The experiment implemented in the ifo Education Survey provides respondents in the treatment group with personalized information about the automatability of their occupation. The experimental setup is as follows (see also Appendix Figure A1): first, we elicit respondents' current occupation and their beliefs about the automatability of their occupation (section 3.3.2). Next, we provide respondents in the treatment group with information about their occupations' automatability (section 3.3.1). Then, we elicit the main outcomes, i.e., labormarket expectations, likelihood of participating in further training and retraining, and the wage fraction they are willing to forgo during further training (section 3.3.3). Respondents in the control group answer the same questions without receiving the information. Finally, all respondents are asked about potential barriers to participating in further training.

⁷ We define structural change to respondents in the following way: *By structural change, we mean the constant transformation of economic sectors accelerated by digital technologies, among other things.*

3.3.1. Information Treatment

We hypothesize that underestimation of the automatability of tasks contributes to low participation rates in further training for some subpopulations of workers. At the beginning of the survey, we ask respondents which occupation they currently work in.⁸ According to respondents' occupations, the treatment provides personalized information about the automatability of the respective occupation to the randomly selected treatment group. We use occupation-level information on automatability from the "IAB-Job-Futuromat" (https://job-futuromat.iab.de/), provided by the Research Institute of the Federal Employment Agency (IAB).⁹ The automatability ranges from zero to 100 percent. The information that we provide reads as follows: *According to a study, [X] percent of core tasks in occupation [answer from earlier question about current occupation] are as of today automatable.* The values in brackets are adjusted for each respondent according to his or her current occupation.¹⁰ Along with the verbal statement, we also provide respondents with a graphical visualization of the information (see Appendix Figure A2). Previous research shows that this measure of automatability has important predictive power for employment growth (Dengler and Matthes 2018b).¹¹

3.3.2. Eliciting Beliefs About Automatability

We elicit respondents' perceptions of the automatability of (i) their current occupation in general and (ii) their own job. Prior to the information treatment, we first ask the following

⁸ Respondents can choose their occupation from a list of more than 4,000 occupations included in BERUFENET, an expert database for training and job descriptions from the German Federal Employment Agency. It is similar to the U.S. Occupational Information Network (O*NET).

⁹ In order to obtain this measure of automatability, the IAB relies on a list of core tasks that are typically done by workers in each occupation. Subsequently, experts code for each of these tasks whether they could be fully automated with currently available technology (for details, see Dengler and Matthes (2018a) and https://job-futuromat.iab.de/faq.html). The "IAB-Job-Futuromat" data, reflecting technological capabilities as of 2019, align closely with our data collection period in May/June 2022. Notably, the widespread availability of large language models, such as ChatGPT, only began after our field phase, with its launch in November 2022. It is important to note that the concept of automatability is dynamic and will evolve with future technological innovations.

¹⁰ For the treatment, we use information on general occupation-specific automatability, which is computed based on the assumption that workers in an occupation devote equal time to all core tasks. This assumption could lead to overestimating or underestimating an individual worker's automatability, contingent upon the actual time they spend on each core task (Dengler and Matthes, 2018b). Importantly, the information treatment explicitly clarifies that it provides insights into the automatability of the *occupation* as a whole, rather than the respondents' individual *jobs*.

¹¹ The approach by Dengler and Matthes (2018a, 2018b) is similar to the job-level (task-based) approach by Arntz et al. (2017), who estimate that about twelve percent of workers in Germany have an automation risk greater than 70 percent (Arntz et al. 2016). For the U.S., Arntz et al. (2016) calculate that approximately nine percent of workers work in jobs with an automation risk above 70 percent, which is lower than the occupation-level approach by Frey and Osborne (2017), who estimate that about 47 percent of jobs are at risk of automation.

question about the share of core tasks respondents think are automatable in their current occupation: *What do you think is the percentage of core activities that people perform in the profession [answer from earlier question about current occupation] that can be automated?* We also provide respondents with an example illustrating the calculation of the automatable share of core tasks, accessible by clicking an icon on the screen. In addition, we elicit respondents' confidence in their beliefs on a seven-point Likert scale.

Second, after providing information to the treatment group, we elicit all respondents' beliefs about the share of automatable core tasks that they perform themselves in their jobs: *What percentage of the core activities you specifically perform in your job do you think can be automated?* We intentionally worded this question differently from the preceding belief-related question to minimize the risk of confusing respondents by asking the exact same question twice, which could potentially bias their responses. By eliciting beliefs about the automatability of both their occupation on average and their own job, we can identify (i) the extent to which respondents are misinformed about the occupation-level data provided in the treatment, and (ii) how correcting these perceptions affects their beliefs about personal automation risk, which ultimately influences their labor-market concerns and willingness to train.¹²

3.3.3. Eliciting Labor-Market and Further-Training Outcomes

Our main outcomes are (i) respondents' labor-market expectations, (ii) respondents' stated likelihood that they will participate in further training and retraining, and (iii) the fraction of wages that respondents are willing to forgo to participate in further training.

First, we elicit respondents' labor-market expectations by asking to what extent they agree with a number of statements regarding their professional future. The nine statements are grouped into two domains: the first measures whether respondents are concerned about their professional future and about being replaced by computers or machines (*labor-market concerns*). The second measures whether respondents expect changes in their work environment in the future, such as changes in the task they perform or the hours they work (*work-environment change*). In particular, respondents stated their agreement to the following statements on a five-point Likert scale: *1*) *I am concerned about my professional future*. *2*) *I*

¹² Note that beliefs regarding the average automatability within an occupation might not align with beliefs about the automatability of a respondent's specific job, since task profiles within a given occupation could vary widely across individual jobs. Furthermore, respondents have unique insights into the particular tasks they undertake in their current role, which may produce differences in their automatability beliefs about their jobs compared to their occupations' average.

will have different tasks in my job in the future than I have now. 3) I have a low risk of becoming unemployed. 4) I am concerned that new technologies will replace many tasks in my job. 5) I believe that my job will no longer exist in a few years. 6) I expect to be paid a higher wage in the future. 7) I will work on more demanding tasks in the future. 8) I will work fewer hours in the future than I do now because computers and computer-controlled machines will replace some of my activities. 9) I will work a lot with computers or computer-controlled machines in the future. We combine items by domain into two indices which are standardized to mean zero and standard deviation one. The index *labor-market concerns* (work-environment change) combines statements 1, 3-6 (statements 2, 7-9), with higher values reflecting greater concerns (greater expected changes).¹³

Second, we gauge respondents' likelihood of engaging in further training (courses designed to enhance skills, typically for their current job) and retraining (courses intended for transitioning to another occupation) within the next two years, on a scale ranging from zero to 100 percent. Further training courses are very heterogeneous in intensity. For the purpose of our study, we follow funding eligibility criteria from the Federal Employment Agency and focus on courses entailing at least 120 hours of training (see section 3.1 for details on the institutional background). In the retraining question, we define it as a program where respondents gain skills for a new occupation, as opposed to enhancing skills for their current job. Besides assessing the likelihood of enrolling in a retraining program, we also inquire which occupation respondents would choose for retraining if they were to do so within the next two years.

Third, to elicit respondents' willingness to pay for further training, we ask them about the fraction of their wages that they would be willing to forgo while completing further training of at least 120 hours outside their company. To answer this question, respondents can indicate a number from zero to 100, where zero indicates that respondents would not be willing to forgo any part of their current wage.

3.4. The Econometric Model

We estimate the effect of the information treatment on outcomes with the following regression model:

$$y_i = \alpha_0 + \alpha_1 Information_i + \delta' X_i + \varepsilon_i$$
(1)

¹³ Combining the nine items into two indices can alleviate concerns of multiple hypothesis testing and improve statistical power (Anderson 2008; Heller et al. 2017).

where y_i is the outcome variable of interest for respondent *i*, e.g., labor-market expectations or likelihood of participating in further training. *Information_i* indicates whether respondent *i* was in the information-treatment group, or the control group. X_i is a vector of control variables, and ε_i is the error term. Since ε_i is uncorrelated with treatment status through randomization, the coefficient α_1 provides an unbiased estimate for the causal treatment effect of information provision even without adding further control variables. As the inclusion of control variables can increase the precision of estimates, we show results with control variables in our analyses.¹⁴

As previously discussed and supported by existing research, individuals in occupations with high automatability are strongly underrepresented in continued education and further training initiatives (Heß et al. 2019; OECD 2021), despite the fact that they would benefit most from further training. Consequently, examining how information provision influences various worker subgroups categorized by their occupations' automatability is particularly interesting from a distributional viewpoint. Therefore, we estimate a second model which includes an interaction term of the treatment indicator and a dummy variable Low Automation_i, coded one if the automatability of a respondent's occupation is less than 50 percent:¹⁵

 $y_i = \alpha_0 + \alpha_1 \text{Information}_i + \alpha_3 \text{Low Automation}_i + \alpha_4 \text{Information}_i \times$

Low Automation_{*i*} + $\delta' X_i + \varepsilon_i$ (2)

We correct for multiple hypothesis testing for our pre-registered primary outcomes, applying the correction proposed by List et al. (2019), which is based on Romano and Wolf (2010). Overall, correcting for multiple outcomes does not change the interpretation of our results. We report corrected p-values in the table notes of Tables 2, 3, and 4.

3.5. Sample Balance and Descriptive Statistics

We perform a balancing test to check whether the randomization worked as intended, i.e., whether respondents' observable characteristics are balanced between treatment and control group (see Appendix Table A1). Reassuringly, only two out of 28 pairwise comparisons are

¹⁴ All qualitative results hold without including control variables (available on request).

¹⁵ We would expect the size of the treatment effects to also correlate with the absolute distance between respondents' prior beliefs and the information treatment. In our sample, we do not have enough power to detect these effects although results show the expected pattern (results available on request).

statistically significant at the five percent level. In addition, item non-response is not correlated with treatment status (see Appendix Table A2).¹⁶

Respondents in our sample work in 1,118 different occupations. The most common occupations are "Management Assistant - Office Management", "Office Clerk" and "Bank Clerk". This corresponds well to administrative data from the Federal Employment Agency demonstrating that most respondents in Germany work in the occupational group "office and secretariat" (Statistik der Bundesagentur für Arbeit 2023b). The average automatability across all occupations in our sample is 51.3 percent. Occupations with the highest automatability (100 percent) are, for example, "Administrative employee" and "Machine, plant, and container cleaner". Occupations with the lowest automatability (zero percent) are occupations such as "Social Worker / Social Pedagogue" and "Care Worker / Everyday Companion".

We also match respondents' reported occupations to additional data on the typical requirement level and the occupational field using the German Classification of Occupations from 2010 (Paulus and Matthes 2013). This classification divides occupations into four requirement-level categories, distinguishing between unskilled or semi-skilled activities, specialist activities, complex specialist activities, and highly complex activities. We compare the distribution of requirement levels in our sample with administrative data from the Federal Employment Agency (Statistik der Bundesagentur für Arbeit 2023a). As it turns out, most respondents in our sample (in administrative data), 52.3 percent (57.3 percent), work in specialist activities, and only 6.1 percent (15.7 percent) in unskilled or semi-skilled activities. 22.9 percent (13.1 percent) work in complex specialist activities, while 18.7 percent (13.8 percent) work in highly complex activities. Consequently, our data exhibits a slight overrepresentation of respondents engaged in complex and highly complex tasks, compared to those in unskilled, semi-skilled, or specialist roles. Given that lower-skilled occupations often have higher automatability, our results may present lower-bound estimates of the information effects for the general population.

In terms of occupational fields, our sample encompasses all major occupations of the German population (see Appendix Table A3). The majority of respondents works in administration and organization occupations (24.2 percent), whereas the smallest proportion works in agriculture (1.6 percent). Compared to the German population, workers in production

¹⁶ In our preferred specification, we do not constrain respondents to having valid answers for all items of the survey. Due to item non-response, this means that observation numbers vary slightly across different specifications. Results remain virtually unchanged if we restrict the sample to respondents who answered all questions. Details available upon request.

and manufacturing are underrepresented in our sample (14.2 percent vs. 20.7 percent in the population). By contrast, workers in commercial services are slightly overrepresented in our sample (16.4 percent vs. 11.4 percent). Overall, our sample, designed to represent the German adult population, accurately mirrors occupation types, fields, and skill levels of workers in Germany.

4. Results

We first present descriptive evidence on respondents' prior beliefs (section 4.1.1) and correlations between prior beliefs, the occupations' actual automatability, and respondents' likelihood of participating in further training (section 4.1.2). Section 4.2 presents our experimental results. Section 4.3 presents descriptive evidence on the reasons for (not) participating in further training.

4.1. Descriptive Results

4.1.1. Are Respondents Aware of Their Occupations' Automatability?

First, we examine whether respondents are aware of their occupations' automatability. Figure 1 shows the distribution of the actual automatability of our respondents' occupations based on data from the "IAB-Job-Futuromat" (transparent bars), and respondents' prior beliefs about their occupations' automatability (blue bars). Occupations' automatability is distributed fairly evenly from zero to 100 percent, while respondents' beliefs are skewed towards lower automatability.

In Figure 2, we plot respondents' beliefs against the actual automatability of their occupations. Those positioned on the 45-degree line have beliefs that correspond with their occupations' actual automatability. Respondents below (above) the 45-degree line underestimate (overestimate) their occupations' automatability. A majority of 67.5 percent of respondents underestimate their automatability, while 21.4 percent overestimate it. Only 11.1 percent are within the accuracy range (i.e., within a five-percentage point deviation above or below the 45-degree line).¹⁷ The correlation between respondents' beliefs and their

¹⁷ Note that rounding is an important concern in the elicitation of continuous beliefs in surveys (Manski and Molinari 2010). We can assess the prevalence of rounding in our context by identifying respondents' rounding behavior throughout the questionnaire. In total, we use four questions in the experiment where respondents indicate continuous answers between zero and 100 with a slider. Only nine percent of respondents indicate a number which is a multiple of ten for each of the four sliders. This suggests that rounding is not a major concern in our study.

occupations' automatability is statistically significantly positive but very small at 0.2 (Figure 2). This implies that respondents in occupations with high automatability have the largest misperceptions of their automatability. Appendix Figure A3 emphasizes that this is driven by differences in automatability, not differences in beliefs: both groups of respondents (with high and low automatability) indicate similar beliefs about their occupations' automatability. Comparing the medians of beliefs and actual automatability, the difference is only approximately 5 percentage points for respondents with low automatability (20.0 vs. 25.0 percent) while it is large at 45 percentage points for respondents with high automatability (30.0 vs. 75.0 percent).

Overall, we document sizeable underestimation of automatability. This is in line with the literature documenting favorable misperceptions in other labor-market beliefs, for example, optimistic bias in job seekers' beliefs about their job-finding probability (Mueller et al. 2021) or underestimation of earnings at another potential employer (Jäger et al. 2021). However, on average, respondents in occupations with higher automatability are significantly less confident about their answer (mean values 4.6 vs. 4.3 on a seven-point scale).

4.1.2. Which Respondents Are Willing to Participate in Further Training?

In this section, we explore whether the willingness to participate in further training and retraining correlates with the automatability of respondents' occupations or their beliefs.

In the control group, respondents state an average likelihood that they will participate in further training (retraining) of 40.7 percent (27.1 percent). The willingness to participate in further training declines slightly with increasing automatability of respondents' occupations. Respondents in occupations with an automatability of below 50 percent state on average a 44.1 percent likelihood of participating in further training, which is 6.5 percentage points higher compared to respondents in high-automatability occupations.¹⁸ Regarding retraining, respondents in occupations with low automatability state a 28.8 percent likelihood of participating of participating with high automatability state a likelihood of 25.6 percent (p < 0.05). This corroborates that the lower participation in further training for

¹⁸ This pattern is also reflected in respondents' previous participation in further training. Respondents with high automatability of their occupation participated in fewer further-training measures in the past than those with low automatability (Appendix Figure A4). Furthermore, respondents in high-automatability occupations who did not previously participate in any training state the lowest likelihood of participating in the future (32.1 percent) compared to those who previously participated (43.3 percent). Respondents in low-automatability occupations without (with) previous training participation state a 34.8 (47.6) percent likelihood of future training participating. These two differences in future training participation by past training participation are statistically significant.

respondents with a high share of routine and automatable core tasks documented in the literature already emerges in stated participation expectations (Heß et al. 2019).

As documented in section 4.1.1., a large proportion of respondents underestimates their occupations' automatability. It seems likely that individual training decisions are based on respondents' beliefs about automatability. As Figure 3 shows, the stated likelihood of participating in further training increases significantly with increasing beliefs about automatability level of 50 percent and declines for higher levels.¹⁹ For retraining, we find a statistically significant and positive relationship throughout: the higher the beliefs about the occupations' automatability, the higher the stated likelihood of participating in retraining. Thus, respondents seem to perceive further training as especially useful if their occupations' susceptibility to automation is moderate, while retraining becomes more attractive with increasing beliefs about automatability.

While these analyses are purely descriptive, we next present the results from our experiment, which allows us to test whether providing information about occupations' automatability causally affects respondents' willingness to train.

4.2. Experimental Results

4.2.1. Beliefs About Own Job's Automatability

We first examine whether information about the automatability of respondents' occupations affects their beliefs about the automatability of their current job. To this end, we estimate treatment effects of information provision on respondents' beliefs about the share of automatable tasks that they perform in their job.

Table 1 shows that the treatment increases respondents' beliefs about the automatability of their own job. Since respondents, on average, underestimate the share of automatable tasks in their occupation (see section 4.1.1), this implies that respondents update their beliefs in line with the information provided. On average, respondents in the control group believe that 26.8 percent of core tasks in their current jobs are automatable (see baseline mean in column 1), which is significantly below the actual average automatability for their occupations (52.1 percent). The treatment increases beliefs among all respondents by 5.1 percentage points. Column 3 reports treatment effects on the difference between the actual automatability of

¹⁹ Figure 3 plots the unconditional relationships between respondents' beliefs and their stated likelihood of participating in further training (dark green) and retraining (light green). The results remain similar when controlling for age and gender.

respondents' occupations and their beliefs about their own job's automatability. While controlgroup beliefs about the automatability of their own job are 25.3 percentage points below the actual automatability of their occupations, the treatment reduces this gap by 7.0 percentage points.

Appendix Figure A5 shows the correlation between respondents' beliefs about their own jobs' automatability and their occupations' automatability, separately for control and treatment group. In the control group, beliefs about their own jobs' automatability are largely flat across the whole spectrum of actual automatability. In the treatment group, respondents' beliefs increase with actual automatability.

The overall information-treatment effect on automatability beliefs is largely driven by respondents in high-automatability occupations. Based on equation (2), the regressions in columns 2 and 4 of Table 1 report treatment effects separately for respondents in occupations with high and low automatability. In the control group, respondents in high-automatability occupations – defined as occupations with an automatability of 50 percent and more - believe on average that their own job's automatability is 30.3 percent, only 7.3 percentage points higher than control-group beliefs of respondents in occupations with low automatability. The treatment increases beliefs among those in high-automatability occupations significantly by 12.9 percentage points, and slightly decreases beliefs among those in low-automatability occupations by 2.4 percentage points (column 2). The treatment effect for respondents in high-automatability occupations in high-automatability and their beliefs about their jobs' automatability and their beliefs about their jobs' automatability and their beliefs about their jobs' automatability is uncomatability and their beliefs about their jobs' automatability (column 4). Thus, the information treatment significantly corrects these respondents' beliefs on their own jobs' automatability upwards.

4.2.2. Labor-Market Expectations

Next, we show that information about occupations' automatability increases respondents' labor-market concerns and affects their expectations about future changes in their work environment. Table 2 presents estimations based on equation (1) in columns 1 and 3, and equation (2) in columns 2 and 4, with the two indices of *labor-market concerns* and *work-environment change* as dependent variables.

On average, the information treatment significantly increases the index of *labor-market concerns* by 9.7 percent of a standard deviation. This overall effect is primarily driven by respondents in occupations with high automatability (column 2): the treatment effect for this

group is 15.9 percent of a standard deviation, while the treatment effect for those in occupations with low automatability is small and statistically insignificant.

Treated respondents are also more likely to expect changes in their future work environment. The treatment increases the index *work-environment change* by 13.0 percent of a standard deviation. Again, the treatment effect is larger for respondents in high-automatability occupations than low-automatability occupations (16.0 versus 10.5 percent of a standard deviation), although the difference between groups is not statistically significant (column 4).

As detailed in the table note, average treatment effects, and treatment effects for respondents in high-automatability occupations remain statistically significant when correcting for multiple hypothesis testing. The interaction terms are no longer statistically significant.

Appendix Table A4 demonstrates that the treatment effects on the indices are due to impacts on various components of these indices.²⁰ The treatment effect on the *labor-market concerns* index of high-automatability respondents is mainly driven by increased concerns that new technologies will replace many tasks in their job (column 3), and that their job will no longer exist in a few years (column 4). The effect on the *work-environment change* index is driven by all index components, but beliefs that respondents will work on more demanding tasks in the future, which is not affected by the treatment. Thus, respondents have a nuanced interpretation of the information on automatability, incorporating both labor-displacing and labor-reinforcing narratives of technological change.

4.2.3. Participation in Further Training and Retraining

We now show that providing information about occupations' automatability also increases respondents' willingness to participate in further training and retraining. In the control group, respondents state a 40.7 percent likelihood of participating in further training on average, and a 27.1 percent likelihood of retraining (Table 3). The information treatment significantly increases the likelihood of participation in further training (retraining) by 2.3 percentage points (3.4 percentage points). These effects correspond to 5.8 and 12.5 percent of the mean, respectively. Appendix Figure A6 reports the distribution of reported likelihoods of participation for the control group (blue bars) and the treatment group (transparent bars). For further training (Panel (a)), the information treatment tends to reduce the share of respondents who report a probability of participation close to zero. For retraining (Panel (b)), the share

²⁰ Moreover, the table shows that only minorities of respondents in the control group are concerned about their professional future, whereas majorities expect to work on more demanding tasks, and work a lot with computers in the future (see bottom of the table).

reporting a zero probability to participate is much lower, and the share reporting a 100 percent probability to participate is much higher. The treatment significantly increases the latter share.

The information-treatment effect on the likelihood to participate in further training and retraining is more pronounced among respondents in high-automatability occupations: the treatment significantly increases their likelihood to participate in further training and retraining by 4.6 and 5.2 percentage points, respectively (see Table 3, columns 2 and 4). The treatment effect for respondents in occupations with low automatability is significantly smaller, and not statistically significant for either outcome.²¹ As discussed in section 4.1.2, willingness to participate in further training and retraining is lower among respondents in high-automatability occupations. The information treatment closes these gaps (almost) entirely.

Consistent with the results on stated likelihood to participate in training, we also find significant treatment effects on respondents' willingness to accept a reduced wage while participating in further training. Table 4 shows that the information treatment significantly increases the fraction of wages respondents are willing to forgo by 1.3 percentage points, or 13.1 percent compared to the control-group mean of 9.6 percent (column 1). Again, this effect is driven by those in occupations with high automatability, for whom the effect is highly statistically significant at 2.6 percentage points (column 2). In addition, the treatment marginally significantly increases their probability to forgo any positive share of their wages (column 4).²²

4.2.4. Further Outcomes

Occupation Choice.²³ We also ask respondents which occupations they would choose for retraining if they were to retrain in the next two years. On average, respondents in the treatment group report retraining occupations with a slightly higher automatability, although this effect is not statistically significant (see Table 5, column 1). One possible explanation for this small effect is that respondents remain uninformed about the automatability of other occupations, even though they receive automatability information about their current occupation. Given the large misperceptions of respondents' own occupation documented in section 4.1.1, it is

²¹ The average treatment effects, and the treatment effects for respondents in high-automatability occupations, remain statistically significant when correcting for multiple hypothesis testing. The interaction terms lose their statistical significance (see notes of Table 3).

²² The average treatment effect as well as the treatment effect for high-automatability respondents remain statistically significant when correcting for multiple hypothesis testing. The effects on the share of those willing to forgo any positive share of their wage become insignificant (see notes of Table 4).

²³ While we pre-registered the question on which retraining occupations respondents would choose, analyses in this section were not pre-registered and are explorative.

plausible that respondents are also unaware of the automatability of other occupations. We also find no statistically significant effect among respondents in high-automatability occupations.

Furthermore, we also find no significant treatment effects on whether respondents want to retrain to (i) another occupational field, (ii) another occupational field with low automatability, or (iii) an occupation with higher requirements than their current occupation (see columns 3-8). However, treated respondents in high-automatability occupations are marginally significantly more likely to retrain to occupations with lower wages (column 10).

Taken together, this analysis suggests that even though treated respondents are more willing to retrain, they do not plan to switch to occupations with lower automatability or higher wages. One interpretation of this finding is that respondents are generally not aware of key characteristics of their occupational outside options (e.g., Jäger et al. 2021).

Policy Views. In this section, we explore how information about occupations' automatability affects respondents' views on public policies related to increase take-up of further training.

First, respondents were asked whether they favor or oppose the policy proposal to make training participation compulsory for persons whose occupation is affected by structural change and digitalization. While 62.5 percent in the control group (rather or strongly) support this proposal, the information treatment does affect these preferences (see Appendix Table A5, columns 1 and 2). We also ask respondents whether they believe that participation in further training is a good strategy to cope with structural change. Three-quarters of respondents (76.7 percent) in the control group agree with this statement, but again the information treatment does not affect this belief (see Appendix Table A5, columns 3 and 4).

Similarly, treated respondents are not more likely to think that the need for further training for (1) all employees in Germany or (2) employees in the same occupation as themselves will increase.²⁴ In the control group, 66.2 percent (50.4 percent) believe that the need for further training for all employees (employees in the same occupation) will "strongly" or "rather" increase (see Appendix Table A5, columns 5 and 7). Hence, respondents perceive a greater need for further training among those in other occupations than their own, which could be due to optimistic biases previously documented for other labor-market decisions (e.g., Mueller et al. 2021).²⁵

²⁴ Questions are worded as follows: What do you think, will the need for further training for the following groups of people increase, decrease, or remain unchanged in the future? (1) For all employees in Germany (2) For people who are in the same occupation as me.

²⁵ Interestingly, this pattern is consistent with the observation that 53.8 percent of Germans think that there are more losers than winners because of structural change, while only 26.6 percent see themselves as a loser of structural change Werner et al. 2022.

Overall, these results show that respondents have a positive view about further training as a means to keep pace with structural change, and support according policy proposals. Despite its effects on respondents' labor-market concerns, the treatment does not affect respondents' policy views, which is a pattern in line with previous research on other policy domains (e.g., Alesina et al. 2018; Kuziemko et al. 2015; Haaland and Roth 2023).

4.3. Reasons for (Non-)Participation

Finally, we ask all respondents about their reasons for (not) wanting to participate in further training. Respondents are asked to what extent they agree with a set of statements relating to financial, time, and employer constraints. Previous literature has identified these as potential barriers to participating in further training (see, e.g., van den Berg et al. 2019, Müller and Wenzelmann 2020, Osiander and Stephan 2020). As shown in Appendix Table A6, almost half of the respondents state that they face financial and employer constraints (45.4 and 45.0 percent, respectively), 35.2 percent face time constraints, and 39.5 percent do not want to participate in measures offered by the Federal Employment Agency. As expected, providing information about automatability does not affect agreement with resource and personal constraints, as these factors are unaffected by the provided information.²⁶

In contrast, respondents' do not seem to feel uninformed about further training possibilities and finance options. Upon being offered extra information at the end of the survey, merely 38.6 percent of respondents choose to access it. The information treatment does not affect information-acquisition behavior (see Appendix Table A8).²⁷

Finally, we examine whether training participation depends on respondents' age and educational attainment.²⁸ One might expect that workers closer to their retirement age are less willing to participate in further training, and react less strongly to the information provided (e.g., Innocenti and Golin 2022). In fact, Table 6 shows that information-treatment effects on willingness to (re)train are confined to respondents below age 60, whereas effects for those above 60 years are small and statistically insignificant (columns 1 and2). A likely reason is that younger workers, with more time before retirement, see greater potential for technological

²⁶ Similarly, we find no information effects on other possible barriers to training, like insecurity about the returns to training, perceived necessity of further training, or confidence about professional future (see Appendix Table A7).

²⁷ We ask respondents the following question: *Would you like to receive more information about further training opportunities, funding, and providers in Germany?*

²⁸ This heterogeneity analysis was not pre-registered and should therefore be viewed as exploratory.

transformation impacting their careers and for recouping the benefits of new skills, unlike older workers who are closer to Germany's retirement age of 67.

Turning to educational attainment, columns 3 and 4 of Table 6 reveal that the information treatment only increases willingness to (re)train among the lower educated (i.e., without university entrance qualification), whereas the higher educated do not react to the treatment. This result is in line with our main finding that information effects are stronger among respondents in high-automatability occupations, reflecting the negative correlation between respondents' educational attainment and automatability.

5. Conclusion

Technological and structural change in the labor market are increasing the demand for new skills in the workforce. Further training and retraining therefore emerge as key elements to bridge the gap between workers' initial education and current developments in skill demand. Yet, participation rates in training programs are especially low for individuals in occupations with high automation risk. We show that one potential reason for the lack of training participation in this vulnerable group of workers are misperceptions regarding the automatability of tasks commonly performed in their occupations. On average, respondents underestimate the automatability of their occupation, especially those in occupations with high automatability. Providing information about the automatability of workers' occupations increases different dimensions of their labor-market concerns, and their beliefs that their work environment will change in the future towards othertasks. Respondents exhibit a nuanced understanding of the provided information regarding automatability, encompassing both labordisplacing and labor-reinforcing narratives of technological change. Importantly, the provided information increases willingness to participate in further training and retraining, and the fraction of wages that respondents are willing to forgo to participate in further training. Treatment effects are larger for respondents in occupations with high automatability, who have the highest risk of experiencing adverse impacts of technological transformation. Thus, providing information about occupations' automatability can reduce inequality in training participation between workers in high- and low-automatability occupations. For future research, it would be interesting to investigate whether the increased willingness to train actually results in increased training participation.

References

Acemoglu, Daron (2009): Introduction to Modern Economic Growth: Princeton University Press.

Acemoglu, Daron; Restrepo, Pascual (2018): The Race between Man and Machine: Implications of Technology for Growth, Factor Shares, and Employment. In *American Economic Review* 108 (6), pp. 1488–1542. DOI: 10.1257/aer.20160696.

Acemoglu, Daron; Restrepo, Pascual (2019): Automation and New Tasks: How Technology Displaces and Reinstates Labor. In *Journal of Economic Perspectives* 33 (2), pp. 3–30. DOI: 10.1257/jep.33.2.3.

Acemoglu, Daron; Restrepo, Pascual (2020a): Robots and Jobs: Evidence from US Labor Markets. In *Journal of Political Economy* 128 (6). DOI: 10.1086/705716.

Acemoglu, Daron; Restrepo, Pascual (2020b): Unpacking Skill Bias: Automation and New Tasks. In *AEA Papers and Proceedings* 110, pp. 356–361. DOI: 10.1257/pandp.20201063.

Acemoglu, Daron; Restrepo, Pascual (2022): Tasks, Automation, and the Rise in U.S. Wage Inequality. In *ECTA* 90 (5), pp. 1973–2016. DOI: 10.3982/ECTA19815.

Aghion, Philippe; Howitt, Peter (1992): A Model of Growth Through Creative Destruction. In *Econometrica* 60 (2), pp. 323–351. DOI: 10.2307/2951599.

Alesina, Alberto; Stantcheva, Stefanie; Teso, Edoardo (2018): Intergenerational Mobility and Preferences for Redistribution. In *American Economic Review* 108 (2), pp. 521–554. DOI: 10.1257/aer.20162015.

Anderson, Michael L. (2008): Multiple Inference and Gender Differences in the Effects of Early Intervention: A Reevaluation of the Abecedarian, Perry Preschool, and Early Training Projects. In *Journal of the American Statistical Association* 103 (484), pp. 1481–1495. DOI: 10.1198/016214508000000841.

Arntz, Melanie; Blesse, Sebastian; Doerrenberg, Philipp (2022): The End of Work is Near, Isn't It? Survey Evidence on Automation Angst. In *ZEW - Centre for European Economic Research Discussion Paper* 22 (036). DOI: 10.2139/ssrn.4247981.

Arntz, Melanie; Gregory, Terry; Zierahn, Ulrich (2016): The Risk of Automation for Jobs in OECD Countries: A Comparative Analysis. In *OECD Social, Employment and Migration Working Papers* 189. DOI: 10.1787/5jlz9h56dvq7-en.

Arntz, Melanie; Gregory, Terry; Zierahn, Ulrich (2017): Revisiting the risk of automation. In *Economics Letters* 159, pp. 157–160. DOI: 10.1016/j.econlet.2017.07.001.

Arntz, Melanie; Gregory, Terry; Zierahn, Ulrich (2020): Digitalisierung und die Zukunft der Arbeit. In *Wirtschaftsdienst* 100 (S1), pp. 41–47. DOI: 10.1007/s10273-020-2614-6.

Autor, David (2022): The Labor Market Impacts of Technological Change: From Unbridled Enthusiasm to Qualified Optimism to Vast Uncertainty. Cambridge, MA.

Autor, David; Chin, Caroline; Salomons, Anna; Seegmiller, Bryan (2022): New Frontiers: The Origins and Content of New Work, 1940–2018. In *NBER Working Paper Series*. DOI: 10.3386/w30389.

Autor, David; Salomons, Anna (2018): Is Automation Labor Share-Displacing? Productivity Growth, Employment, and the Labor Share. In *Brookings Papers on Economic Activity* 2018 (1), pp. 1–87. DOI: 10.1353/eca.2018.0000.

Autor, David H.; Katz, Lawrence F.; Krueger, Alan B. (1998): Computing Inequality: Have Computers Changed the Labor Market? In *The Quarterly Journal of Economics*, pp. 1169–1213.

Bessen, James (2019): Automation and jobs: When technology boosts employment. In *Economic Policy* (34), pp. 589–626.

Bessen, James; Goos, Maarten; Salomons, Anna; van den Berge, Wiljan (2020): Automation: A Guide for Policymakers. In *Economic Studies at Brookings*.

Blanas, Sotiris; Gancia, Gino; Lee, Sang Yoon (2019): Who is afraid of machines? In *Economic Policy* 34 (100), pp. 627–690. DOI: 10.1093/epolic/eiaa005.

BMBF (2018): Weiterbildungsverhalten in Deutschland 2018. Ergebnisse des Adult Education Survey - AES-Trendbericht.

Bundesinstitut für Berufsbildung (Ed.) (2020): Datenreport zum Berufsbildungsbericht 2020. Informationen und Analysen zur Entwicklung der beruflichen Bildung.

Bundesministerium für Arbeit und Soziales (Ed.) (2021): Forschungsbericht 526/3. Aktualisierte BMAS-Prognose.

Bursztyn, Leonardo (2016): Poverty and the Political Economy of Public Education Spending: Evidence from Brazil. In *Journal of the European Economic Association* 14 (5), pp. 1101–1128. DOI: 10.1111/jeea.12174.

Cortes, Guido Matias (2016): Where Have the Middle-Wage Workers Gone? A Study of Polarization Using Panel Data. In *Journal of Labor Economics* 34 (1), pp. 63–105. DOI: 10.1086/682289.

Cruces, Guillermo; Perez-Truglia, Ricardo; Tetaz, Martin (2013): Biased perceptions of income distribution and preferences for redistribution: Evidence from a survey experiment. In *Journal of Public Economics* 98, pp. 100–112. DOI: 10.1016/j.jpubeco.2012.10.009.

Dachs, Bernhard (2018): The impact of new technologies on the labour market and the social economy. Study. Brussels: EPRS. Available online at

https://publications.europa.eu/de/publication-detail/-/publication/0adbed9c-8bca-11e8-8a53-01aa75ed71a1/language-en/format-PDF/source-73929611.

Dauth, Wolfgang; Findeisen, Sebastian; Suedekum, Jens; Woessner, Nicole (2021): The Adjustment of Labor Markets to Robots. In *Journal of the European Economic Association* 19 (6), pp. 3104–3153. DOI: 10.1093/jeea/jvab012.

Dengler, Katharina; Matthes, Britta (2018a): Substituierbarkeitspotenziale von Berufen: Wenige Berufsbilder halten mit der Digitalisierung Schritt. In *IAB Kurzbericht* (4).

Dengler, Katharina; Matthes, Britta (2018b): The impacts of digital transformation on the labour market: Substitution potentials of occupations in Germany. In *Technological Forecasting and Social Change* 127, pp. 304–316. DOI: 10.1016/j.techfore.2018.09.024.

Freundl, Vera; Grewenig, Elisabeth; Kugler, Franziska; Lergetporer, Philipp; Schüler, Ruth; Wedel, Katharina et al. (2022): The ifo Education Survey 2014–2021: A New Dataset on Public Preferences for Education Policy in Germany. In *Jahrbücher für Nationalökonomie und Statistik* 0 (0). DOI: 10.1515/jbnst-2022-0051.

Frey, Carl Benedikt; Osborne, Michael A. (2017): The future of employment: How susceptible are jobs to computerisation? In *Technological Forecasting and Social Change* 114, pp. 254–280. DOI: 10.1016/j.techfore.2016.08.019.

Georgieff, Alexandre; Milanez, Anna (2021): What happened to jobs at high risk of automation? In *OECD Social, Employment and Migration Working Papers* 255. DOI: 10.1787/10bc97f4-en.

Golin, Marta; Rauh, Christopher (2022): The Impact of Fear of Automation. In *Cambridge Working Papers In Economics*.

Graetz, Georg; Michaels, Guy (2018): Robots at Work. In *The Review of Economics and Statistics* 100 (5), pp. 753–768. DOI: 10.1162/rest_a_00754.

Haaland, Ingar; Roth, Christopher (2023): Beliefs about Racial Discrimination and Support for Pro-Black Policies. In *The Review of Economics and Statistics* 105 (1), pp. 40–53. DOI: 10.1162/rest_a_01036.

Heller, Sara B.; Shah, Anuj K.; Guryan, Jonathan; Ludwig, Jens; Mullainathan, Sendhil; Pollack, Harold A. (2017): Thinking, Fast and Slow? Some Field Experiments to Reduce Crime and Dropout in Chicago. In *The Quarterly Journal of Economics* 132 (1), pp. 1–54. DOI: 10.1093/qje/qjw033.

Heß, Pascal; Janssen, Simon; Leber, Ute (2019): Digitalisierung und berufliche Weiterbildung: Beschäftigte, deren Tätigkeiten durch Technologien ersetzbar sind, bilden sich seltener weiter. In *IAB-Kurzbericht* (16).

Innocenti, Stefania; Golin, Marta (2022): Human capital investment and perceived automation risks: Evidence from 16 countries. In *Journal of Economic Behavior & Organization* 195, pp. 27–41. DOI: 10.1016/j.jebo.2021.12.027.

Jäger, Simon; Roth, Christopher; Roussille, Nina; Schoefer, Benjamin (2021): Worker Beliefs About Outside Options. In *NBER Working Paper Series* (29623). DOI: 10.3386/w29623.

Jeffrey, Karen (2021): Automation and the future of work: How rhetoric shapes the response in policy preferences. In *Journal of Economic Behavior & Organization* 192, pp. 417–433. DOI: 10.1016/j.jebo.2021.10.019.

Jones, Charles (Ed.) (2002): Introduction To Economic Growth. 2nd Edition. New York, NY, USA: W.W. Norton.

Katz, Lawrence F.; Murphy, Kevin M. (1992): Changes in Relative Wages, 1963-1987: Supply and Demand Factors. In *Quarterly Journal of Economics* 107 (1), pp. 35–78.

Kuziemko, Ilyana; Norton, Michael I.; Saez, Emmanuel; Stantcheva, Stefanie (2015): How Elastic Are Preferences for Redistribution? Evidence from Randomized Survey Experiments. In *American Economic Review* 105 (4), pp. 1478–1508. DOI: 10.1257/aer.20130360.

List, John A.; Shaikh, Azeem M.; Xu, Yang (2019): Multiple hypothesis testing in experimental economics. In *Exp Econ* 22 (4), pp. 773–793. DOI: 10.1007/s10683-018-09597-5.

Manski, Charles F.; Molinari, Francesca (2010): Rounding Probabilistic Expectations in Surveys. In *Journal of Business & Economic Statistics* 28 (2), pp. 219–231. DOI: 10.1198/jbes.2009.08098.

Montobbio, Fabio; Staccioli, Jacopo; Virgillito, Maria; Vivarelli, Marco (2022): The Empirics of Technology, Employment and Occupations: Lessons Learned and Challenges Ahead. In *IZA Discussion Paper Series* (15731). DOI: 10.2139/ssrn.4281286.

Mueller, Andreas I.; Spinnewijn, Johannes; Topa, Giorgio (2021): Job Seekers' Perceptions and Employment Prospects: Heterogeneity, Duration Dependence, and Bias. In *American Economic Review* 111 (1), pp. 324–363. DOI: 10.1257/aer.20190808.

Müller, Normann; Wenzelmann, Felix (2020): Berufliche Weiterbildung – Teilnahme und Abstinenz. In *Zeitschrift für Weiterbildungsforschung* 43 (1), pp. 47–73. DOI: 10.1007/s40955-019-0141-0.

Nedelkoska, Ljubica; Quintini, Glenda (2018): Automation, skills use and training. In *OECD Social, Employment and Migration Working Papers* (202). DOI: 10.1787/2e2f4eea-en.

OECD (1998): Technology, Productivity and Job Creation: Best Policy Practices. 1998 Edition. Paris: OECD Publishing (The OECD Jobs Strategy).

OECD (2021): Continuing Education and Training in Germany. Getting Skills Right. Paris: OECD Publishing.

Osiander, Christopher; Stephan, Gesine (2020): Was beeinflusst die Weiterbildungsbereitschaft von Beschäftigten? Befunde aus einer Vignettenstudie. In *Industrielle Beziehungen* 27 (3), pp. 336–359. DOI: 10.3224/indbez.v27i3.05.

Paulus, Wiebke; Matthes, Britta (2013): The German Classification of Occupations 2010 – Structure, Coding and Conversion Table. In *FDZ-Methodenbericht* (08). Available online at https://doku.iab.de/fdz/reporte/2013/MR_08-13_EN.pdf.

Romano, Joseph P.; Wolf, Michael (2010): Balanced Control of Generalized Error Rates. In *The Annals of Statistics* 38 (1), pp. 598–633. DOI: 10.1214/09-AOS734.

Schmidpeter, Bernhard; Winter-Ebmer, Rudolf (2021): Automation, unemployment, and the role of labor market training. In *European Economic Review* 137 (103808). DOI: 10.1016/j.euroecorev.2021.103808.

Statistik der Bundesagentur für Arbeit (2022): Tabellen, Beschäftigte nach Berufen (KldB 2010) (Quartalszahlen). Edited by Bundesagentur für Arbeit Statistik. Nürnberg.

Statistik der Bundesagentur für Arbeit (2023a): Berufe auf einen Blick. Edited by Statistik der Bundesagentur für Arbeit. Available online at

https://statistik.arbeitsagentur.de/DE/Navigation/Statistiken/Interaktive-Statistiken/Berufeauf-einen-Blick/Berufe-auf-einen-Blick-Anwendung-Nav.html, checked on 6/27/2023.

Statistik der Bundesagentur für Arbeit (2023b): Datenbank Beschäftigung. Edited by Statistik der Bundesagentur für Arbeit. Available online at

https://statistik.arbeitsagentur.de/DE/Navigation/Statistiken/Interaktive-

Statistiken/Datenbanken/Datenbanken-BST-Nav.html, checked on 6/27/2023.

van den Berg, Gerard J.; Dauth, Christine; Homrighausen, Pia; Stephan, Gesine (2019): Informing employees in small and medium sized firms about training: results of a randomized field experiment. In *IAB-Discussion Paper*.

Werner, Katharina; Freundl, Vera; Kugler, Franziska; Lergetporer, Philipp; Wedel, Katharina; Wößmann, Ludger (2022): Deutsche befürworten Weiterbildung, um mit dem Strukturwandel Schritt zu halten. Ergebnisse des ifo Bildungsbarometers 2022. In *ifo Schnelldienst* (9), pp. 56–69.

Figures and Tables

Figures



Figure 1: Distribution of respondents' beliefs and automatability of respondents' occupations

Notes: Blue bars depict bins for respondents' answers to the question "*What do you think is the percentage of core activities that people perform in the profession [answer from earlier question about current occupation] that can be automated?*". Transparent bars depict bins for the automatability of respondents' occupations according to expert estimates by the Research Institute of the Federal Employment Agency in Germany (IAB). Data source: ifo Education Survey 2022.

Figure 2: Difference between respondents' beliefs and occupations' automatability



Notes: Respondents' answers to the question "*What do you think is the percentage of core activities that people perform in the profession [answer from earlier question about current occupation] that can be automated?*" are depicted as averages for each occupation's automatability (calculated by experts from the IAB). The blue line depicts the prediction line and the 95 percent confidence interval. The black line depicts the 45-degree line with a 5-point bandwidth. Points above the 45-degree line indicate an overestimation of the occupation's automatability and points below indicate an underestimation. Data source: ifo Education Survey 2022.





Notes: Respondents' answers to the questions "How likely is it that you yourself will participate in further training of at least 120 hours within the next two years?" and "How likely is it that within the next two years you will complete retraining to another occupation?" are depicted as averages for 10 bins of respondents' beliefs about their occupations' automatability. The dark green dots depict the averages for the likelihood to participate in further training, and the light green dots depict the ones for retraining. Data source: ifo Education Survey 2022.

Tables

	Beliefs about own	job's automatability	Difference occupation's automat autom	ability and beliefs about own job's atability
	(1)	(2)	(3)	(4)
Information	5.124***	12.874***	-6.989***	-13.858***
	(0.879)	(1.260)	(1.123)	(1.336)
Low Automation		-7.299***		-43.694***
		(1.192)		(1.333)
Information x Low Automation		-15.321***		14.958***
		(1.646)		(1.801)
Covariates	Yes	Yes	Yes	Yes
Baseline mean	26.822	30.290	25.290	46.857
Observations	3005	3005	3005	3005
R-squared	0.056	0.168	0.063	0.394
Information Effect for Low Automation		-2.447**		1.100
		(1.065)		(1.213)

Table 1: Effect of information treatment on beliefs

Notes: OLS regressions. Dependent variables: (1) - (2) Beliefs about own job's automatability; (3) - (4) Occupation's actual automatability minus beliefs about own job's automatability; Randomized experimental treatment group "Information": respondents informed about automatability of occupation. Baseline mean: Mean of the variable in the control group or baseline group. Covariates include: Age, female, born in Germany, West Germany, living in large city, risk, patience, parents with university education, income, current employment status, middle school degree, high school degree, partner living in household, parental status, and imputation dummies. Outcomes in this table are pre-registered as secondary outcomes. Data source: ifo Education Survey 2022. Robust standard errors in parentheses. Significance levels: * p < 0.1, ** p < 0.05, *** p < 0.01.

Table 2: Information effect on labor-market expectations (index)

	Index labor-market concerns		Index work-envi	ronment change
	(1)	(2)	(3)	(4)
Information	0.097***	0.159***	0.130***	0.160***
	(0.036)	(0.049)	(0.034)	(0.046)
Low Automation		-0.194***		-0.129***
		(0.049)		(0.048)
Information x Low Automation		-0.118*		-0.055
		(0.071)		(0.067)
Covariates	Yes	Yes	Yes	Yes
Baseline Mean	0.000	0.084	0.000	0.036
Observations	3011	3011	3011	3011
R-squared	0.088	0.103	0.148	0.155
Information Effect for Low Automation		0.041		0.105**
		(0.050)		(0.049)

Notes: OLS regressions. Dependent variables: (1) - (2) Index labor-market concerns; (3) - (4) Index work-environment change; Randomized experimental treatment group "Information": respondents informed about automatability of occupation. Baseline mean: Mean of the variable in the control group or baseline group. See Table 1 for included covariates. Outcomes in this table are pre-registered as primary outcomes. MHT corrected p-values: for Information: 0.032 in column (1), 0.015 in column (2), 0.001 in column (3), 0.001 in column (4). For Information x Low Automation: 0.159 in column (2), 0.414 in column (4). Data source: ifo Education Survey 2022. Robust standard errors in parentheses. Significance levels: * p < 0.1, ** p < 0.05, *** p < 0.01.

	Likelihood further training		Likelihood	retraining
	(1)	(2)	(3)	(4)
Information	2.343**	4.647***	3.401***	5.194***
	(1.063)	(1.465)	(0.999)	(1.389)
Low Automation		4.977***		1.860
		(1.498)		(1.359)
Information x Low Automation		-4.755**		-3.658*
		(2.132)		(2.007)
Covariates	Yes	Yes	Yes	Yes
Baseline Mean	40.740	37.576	27.111	25.555
Observations	2973	2973	2928	2928
R-squared	0.142	0.145	0.188	0.189
Information Effect for Low Automation		-0.108		1.535
		(1.541)		(1.441)

Table 3: Information effect on likelihood to participate in further training and retraining

Notes: OLS regressions. Dependent variables: (1) - (2) Stated likelihood to participate in further training (between zero to 100); (3) - (4) Stated likelihood to participate in retraining (between zero to 100); Randomized experimental treatment group "Information": respondents informed about automatability of occupation. Baseline mean: Mean of the variable in the control group or baseline group. See Table 1 for included covariates. Outcomes in this table are pre-registered as primary outcomes. MHT corrected p-values: for Information: 0.051 in column (1), 0.001 in column (2), 0.001 in column (3), 0.001 in column (4). For Information: 0.120 in column (2), 0.178 in column (4). Data source: ifo Education Survey 2022. Robust standard errors in parentheses. Significance levels: * p < 0.1, ** p < 0.05, *** p < 0.01.

Table 4: Information effect on willingness to forgo wage for training

	Willingness	o forgo wage Willingness to forgo wage > 0		forgo wage > 0
	(1)	(2)	(3)	(4)
Information	1.250**	2.612***	0.014	0.048*
	(0.613)	(0.871)	(0.017)	(0.024)
Low Automation		0.354		0.026
		(0.841)		(0.025)
Information x Low Automation		-2.736**		-0.068*
		(1.225)		(0.035)
Covariates	Yes	Yes	Yes	Yes
Baseline Mean	9.558	8.999	0.476	0.449
Observations	3008	3008	3008	3008
R-squared	0.103	0.105	0.106	0.107
Information Effect for Low Automation		-0.124		-0.020
		(0.864)		(0.025)

Notes: OLS regressions. Dependent variables: (1) - (2) Share of wage willing to forgo (zero to 100); (3) - (4) Dummy variable coded one if stated willingness is non-zero (positive); Randomized experimental treatment group "Information": respondents informed about automatability of occupation. Baseline mean: Mean of the variable in the control group or baseline group. See Table 1 for included covariates. Outcomes in this table are pre-registered as primary outcomes. MHT corrected p-values: for Information: 0.079 in column (1), 0.010 in column (2), 0.397 in column (3), 0.208 in column (4). For Information x Low Automation: 0.137 in column (2), 0.220 in column (4). Data source: ifo Education Survey 2022. Robust standard errors in parentheses. Significance levels: * p < 0.1, ** p < 0.05, *** p < 0.01

Table 5: Characteristics of retraining occupations

					Low au	tomation	Hi	gher		
	Autom	atability	Other occup	pational field	occupat	ional field	requirer	nent level	Mean wage	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)
Information	1.217	1.686	0.010	-0.028	-0.015	-0.035	-0.008	-0.031	-68.423	-117.155*
	(1.237)	(1.736)	(0.019)	(0.027)	(0.018)	(0.024)	(0.019)	(0.028)	(48.633)	(67.486)
Low Automation		-9.578***		-0.089***		0.101***		-0.163***		-123.730*
		(1.728)		(0.028)		(0.026)		(0.027)		(68.479)
Information x Low Automation		-1.099		0.077**		0.040		0.046		98.639
		(2.440)		(0.039)		(0.037)		(0.038)		(97.380)
Covariates	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Baseline Mean	44.024	49.308	0.638	0.684	0.318	0.263	0.348	0.429	3889.858	3897.594
Observations	2326	2326	2449	2449	2449	2449	2447	2447	2190	2190
R-squared	0.027	0.054	0.010	0.014	0.049	0.066	0.017	0.039	0.146	0.147
Information Effect for Low Automation		0.587		0.049*		0.005		0.015		-18.517
		(1.711)		(0.028)		(0.027)		(0.026)		(70.552)

Notes: OLS regressions. Dependent variables: (1) - (2): Automatability of the indicated retraining occupation (zero to 100); (3) - (4): Dummy variable indicating whether the retraining occupation is in a low automation occupational field (health & social services, social science and agriculture); (5) - (6): Dummy variable indicating whether the retraining occupational field; (7) - (8): Dummy variable indicating whether the retraining occupation is of a higher requirement level; (9) - (10): Mean wage of the retraining occupation. Randomized experimental treatment group "Information": respondents informed about automatability of occupation. Baseline mean: Mean of the variable in the control group or baseline group. See Table 1 for included covariates. Data source: ifo Education Survey 2022. Robust standard errors in parentheses. Significance levels: * p < 0.1, ** p < 0.05, *** p < 0.01.

Table 6: Heterogeneity of information treatment effects across subgroups

	Subgroup: A	ge above 60	Subgroup: University Er	ntrance Qualification
	Likelihood further training	Likelihood retraining	Likelihood further training	Likelihood retraining
	(1)	(2)	(3)	(4)
Information [baseline: not in subgroup]	2.575**	3.683***	4.375***	4.667***
	(1.113)	(1.072)	(1.422)	(1.307)
Subgroup	-4.238	-4.838**	5.246***	-1.802
	(2.812)	(2.027)	(1.910)	(1.735)
Information x Subgroup	-2.732	-3.378	-4.734**	-2.955
	(3.698)	(2.578)	(2.139)	(2.026)
Covariates	Yes	Yes	Yes	Yes
Baseline Mean	42.288	28.932	35.349	24.548
Observations	2973	2928	2970	2925
R-squared	0.144	0.191	0.143	0.188
Information Effect for Subgroup	-0.157	0.305	-0.359	1.712
	(3.524)	(2.347)	(1.597)	(1.550)

Notes: OLS regressions. Dependent variables: (1) Stated likelihood to participate in further training (zero to 100); (2) Stated likelihood to participate in retraining (zero to 100); (3) Stated likelihood to participate in further training (zero to 100); (4) Stated likelihood to participate in retraining (zero to 100); Randomized experimental treatment group "Information": respondents informed about automatability of occupation. Baseline mean: Mean of the variable in the control group or baseline group. See Table 1 for included covariates. Heterogeneities in this table are pre-registered. Data source: ifo Education Survey 2022. Robust standard errors in parentheses. Significance levels: * p < 0.1, ** p < 0.05, *** p < 0.01.

Appendix

Appendix A: Institutional Background

This section provides an overview of the further training system in Germany. The system is characterized by a high degree of decentralization due to Germany's federal structure.

There are five types of learning provision: Basic Continuing Education and Training (CET), General CET, Vocational CET, CET in Higher Education and Adult Liberal Education (OECD 2021). Basic CET refers to non-formal learning opportunities for adults lacking basic skills. General CET includes formal education opportunities for adults to obtain school leaving certificates. Vocational CET encompasses formal and non-formal learning opportunities covering different levels, ranging from basic vocational qualifications to Master crafts people, Bachelor's degrees and certified business economists (OECD 2021). It also includes vocational retraining, adjustment training and vocational upskilling. CET in Higher Education includes Bachelor's and Master's degree programs, while Adult Liberal Education comprises learning opportunities offered by Adult Education Centers.

In addition, there are several types of further training: In-company training, individual jobrelated, and non-job-related. In-company training is the most common type. The average duration for in-company further training is 29 hours, individual job-related approx. 153 hours, and non-job-related further training is 56 hours (BMBF 2018).

Overall, there are approximately 18,000 public and private further training providers (Bundesinstitut für Berufsbildung 2020), including public institutions such as vocational schools or higher education institutions, CET institutions operated by enterprises, or groups of enterprises, social and economic partners such as trade unions and employer organizations, Chambers of Commerce and Trade, Chambers of Skilled Crafts, CET institutions run by churches, political parties, trade unions, foundations, other associations and Adult Education Centers.

The further training system in Germany is subject to various levels of regulation and is governed by numerous legal bases. These include collective bargaining agreements and company agreements, laws, and regulations at the state level. Companies, employees, and the public sector share the responsibility and obligation for further vocational training and its funding.

According to the IW Continuing Education Survey 2020, the participation rate in further education by German companies was approximately 88 percent in the year 2019. Furthermore, the ifo Education Survey reported that 63 percent of respondents stated that they had

participated in further training in the past. Conversely, this means that more than one third (37 percent) have not yet participated in any further training (Werner et al. 2022).

Furthermore, the German government provides support for further education through assistance and funding programs. For example, the Federal Employment Agency offers financial support to job seekers who wish to participate in further training measures to enhance their employability. Additionally, there are various government educational grants and tax benefits available to companies that invest in the further education of their employees.

Appendix B: Figures and Tables

Appendix Figure A1: Visual representation of information



Notes: Experimental setup. Questions in bold are primary outcomes as specified in the AEA Registry AEARCTR-0009464.

Appendix Figure A2: Visual representation of information





Notes: Example of the information about the occupation's automatability provided to the treatment group. This graph shows an example for an occupation with 20 percent automatable core tasks.

Appendix Figure A3: Distribution of respondents' beliefs and occupations' automatability for high- and low-automatable occupations



Notes: Blue shapes show the distribution of respondents' prior beliefs about their occupations' automatability, and red shapes show respondents' occupations' automatability according to expert estimates from the IAB, by respondents' occupations' automatability (high vs. low automatability). The box plots within the shapes show the median value and the interquartile range, with the extended lines representing upper- and lower-adjacent values. Data source: ifo Education Survey 2022.



Appendix Figure A4: Previous participation in further training

Notes: Previous participation in further training of respondents, divided into two groups: Those in occupations with high automatability (larger than 50 percent) and those in occupations with low automatability (less than 50 percent). Data source: ifo Education Survey 2022.

Appendix Figure A5: Respondents' beliefs vs. occupations' automatability by treatment group



Notes: Respondents' answers to the question "*What do you think is the percentage of core activities that people perform in the profession [answer from earlier question about current occupation] that can be automated?*" are depicted as averages for each occupation's automatability (calculated by experts from the IAB), by respondents' treatment status. Randomized experimental treatment group: Respondents informed about the automatability of their occupations. The black line depicts the 45-degree line with a 5-point bandwidth. Points above the 45-degree line indicate an overestimation of the occupation's automatability and points below indicate an underestimation. Data source: ifo Education Survey 2022.



Appendix Figure A6: Likelihood of participation in further training and retraining

Panel (b) Retraining

Notes: Respondents' answers to the questions "How likely is it that you yourself will participate in further training of at least 120 hours within the next two years?" and "How likely is it that within the next two years you will complete retraining to another occupation?" are depicted by respondents' treatment status. Randomized experimental treatment group: Respondents informed about the automatability of their occupations. Data source: ifo Education Survey 2022.

Control variable	Mean control	Mean treatment	Diff.	p-value
Age	42.65	42.98	0.33	0.49
Female	0.47	0.46	-0.01	0.67
Born in Germany	0.93	0.92	-0.02	0.12
City	0.37	0.40	0.03	0.11
Partner	0.64	0.65	0.02	0.35
Parent(s) with university degree	0.38	0.38	-0.01	0.77
Highest educational degree				
No degree/basic degree	0.24	0.26	0.02	0.16
Middle school degree	0.32	0.32	0.00	0.88
High school degree	0.44	0.42	-0.02	0.28
Employment status				
Fulltime	0.66	0.66	0.00	0.95
Parttime	0.22	0.23	0.01	0.50
Self-employed	0.07	0.05	-0.02	0.05
Unemployed	0.00	0.00	0.00	0.98
Retired/III/etc.	0.05	0.06	0.01	0.50
Parent status	0.54	0.54	-0.01	0.77
Party preference				
CDU	0.17	0.17	-0.01	0.61
SPD	0.18	0.17	-0.01	0.61
Grüne	0.13	0.12	-0.01	0.55
Linke	0.05	0.06	0.00	0.85
FDP	0.08	0.09	0.00	0.73
AfD	0.09	0.11	0.02	0.03
None	0.27	0.27	-0.01	0.57
Other	0.02	0.02	0.00	0.75
General voting	0.79	0.80	0.01	0.45
Patience	7.04	7.02	-0.02	0.84
Risk	5.72	5.78	0.06	0.51
Monthly Household Income (€)	2944.21	3014.03	69.82	0.26
West Germany	0.81	0.80	-0.01	0.56

Appendix Table A1: Sample Balance

Notes: Group means. 'Diff.' displays the difference in means between the control group and the treatment group who received the information about the automatability of their occupation. Data source: ifo Education survey 2022.

Appendix Table A2: Non-response

	Mean control	Mean treatment	Diff.	p-value
Labor-market expectations				
Concerned future	0.001	0.002	0.001	0.311
Other job tasks	0.000	0.001	0.001	0.157
Low risk unemployment	0.001	0.001	0.001	0.555
Automation tasks	0.000	0.001	0.001	0.317
Occupation existence	0.000	0.001	0.001	0.317
Higher wages	0.000	0.001	0.001	0.317
More demanding tasks	0.000	0.001	0.001	0.317
Less hours	0.000	0.001	0.001	0.157
More computer tasks	0.000	0.001	0.001	0.157
Need further training: all employees	0.001	0.000	-0.001	0.317
Need further training: same job employees	0.001	0.000	-0.001	0.317
Likelihood further training	0.015	0.011	-0.004	0.280
Likelihood retraining	0.029	0.027	-0.002	0.712
Policy proposal: compulsory further training	0.002	0.000	-0.002	0.083
Forgo wage	0.001	0.002	0.001	0.311
Information acquisition	0.000	0.001	0.001	0.157
Financial constraints	0.001	0.001	0.001	0.555
Time constraints	0.000	0.001	0.001	0.157
Employer constraints	0.000	0.001	0.001	0.317
FEA offered further training	0.000	0.001	0.001	0.317
Gains insecure	0.000	0.001	0.001	0.317
Great necessity	0.001	0.001	0.000	0.991
Confidence job future	0.000	0.001	0.001	0.317
Good measure structural change	0.001	0.001	0.000	0.991

Notes: Group means of item non-response. 'Diff.' displays the difference in means between the control group and the treatment group who received the information about the automatability of their occupation. Data source: ifo Education Survey 2022.

Occupational Field	Share of respondents in	Share of German population	Difference
	respective neid	in respective field	
Agriculture	1.59	1.50	0.09
Production & Manufacturing	14.18	20.71	-6.53
Construction & Architecture	4.02	6.10	-2.08
Natural Sciences	6.91	4.29	2.62
Transport	12.22	13.41	-1.19
Commercial Services	16.44	11.38	3.03
Administration & Organization	24.18	20.39	3.79
Health & Social Services	15.84	18.90	-3.06
Social Sciences	4.62	2.78	1.84

Appendix Table A3: Occupational Fields Respondents vs. German Population

Notes: Shares of respondents in respective fields in respondent sample and in German population. Data sources: ifo Education Survey 2022 and Statistik der Bundesagentur für Arbeit 2022.

Appendix Table A4: Labor-Market Expectations

	Index LM concerns					More			
	Concerned future	Low risk unemployment	New tech replaces tasks	Job will not exist	Higher wage	Other tasks in job	demanding tasks	Work less hours	computer tasks
	(1)	(2)	(3)	(4)	(5)	(2)	(7)	(8)	(9)
Information	0.118*	0.006	0.254***	0.224***	0.022	0.246***	0.067	0.177***	0.195***
	(0.068)	(0.068)	(0.067)	(0.065)	(0.064)	(0.065)	(0.060)	(0.065)	(0.069)
Low Automation	-0.069	0.235***	-0.307***	-0.112*	0.158**	-0.128*	-0.085	-0.151**	-0.360***
	(0.069)	(0.069)	(0.066)	(0.064)	(0.065)	(0.066)	(0.062)	(0.065)	(0.071)
Information x Low Automation	-0.091	-0.025	-0.180*	-0.196**	0.055	-0.160*	0.029	-0.090	-0.078
	(0.098)	(0.096)	(0.094)	(0.092)	(0.090)	(0.094)	(0.087)	(0.093)	(0.100)
Covariates	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Baseline Mean Control share rather/fully	2.438	3.458	2.563	2.001	3.269	2.814	3.206	2.291	3.231
agree	0.301	0.649	0.304	0.184	0.585	0.394	0.536	0.253	0.520
Observations	3008	3009	3011	3011	3011	3010	3011	3010	3010
R-squared Information Effect for Low	0.076	0.045	0.086	0.071	0.078	0.089	0.104	0.091	0.096
Automation	0.027	-0.019	0.074	0.028	0.077	0.086	0.096	0.087	0.117
	(0.070)	(0.068)	(0.066)	(0.065)	(0.064)	(0.068)	(0.063)	(0.066)	(0.072)

Notes: OLS regressions. Dependent variables (5-point scale: 1 = fully disagree, 5 = fully agree): (1) I am concerned about my professional future; (2) I have a low risk of becoming unemployed; (3) I am concerned that many tasks in my occupation will be replaced by new technologies; (4) I believe that my occupation will no longer exist in a few years; (5) I expect to be paid a higher wage in the future; (6) I will have different tasks in my occupation in the future than I have now; (7) I will work on more demanding tasks in the future; (8) I will work fewer hours in the future than I do now because some of my activities will be replaced by computers and computer-controlled machines; (9) In the future, I will work a lot with computers and computer-controlled machines. Randomized experimental treatment group "Information": respondents informed about automatability of occupation. Baseline mean: Mean of the variable in the control group or baseline group. Control share rather/fully agree: Dummy variable indicating the share of respondents answering rather or fully agree to the statement. See Table 1 for included covariates. Data source: ifo Education Survey 2022. Robust standard errors in parentheses. Significance levels: * p < 0.1, ** p < 0.05, *** p < 0.01.

Appendix Table A5: Policy views

	Support compulsory further training		Further training is good strategy to cope with structural change		Need for further training will increase (all employees)		Need for further training will increase (same occupation)	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Information	0.011	0.015	0.014	0.009	-0.015	-0.009	0.026	0.031
	(0.017)	(0.024)	(0.015)	(0.021)	(0.017)	(0.024)	(0.018)	(0.025)
Low Automation		0.035		0.015		-0.002		-0.037
		(0.024)		(0.022)		(0.024)		(0.025)
Information x Low Automation		-0.010		0.010		-0.011		-0.010
		(0.034)		(0.030)		(0.034)		(0.036)
Covariates	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Baseline Mean	0.625	0.605	0.767	0.760	0.662	0.654	0.504	0.514
Observations	3009	3009	3010	3010	3011	3011	3011	3011
R-squared	0.059	0.060	0.054	0.054	0.053	0.054	0.051	0.053
Information Effect for Low Automation		0.005		0.019		-0.020		0.021
		(0.024)		(0.021)		(0.024)		(0.026)

Notes: OLS regressions. Dependent variables: (1) - (2) Policy proposal: Obligation for all employees affected by structural change and digitalization to participate in further training (dummy coded one if agree); (3) - (4) Further training is a good way to keep pace with structural change; (5) - (6) Need for further training for all employees (dummy coded one if increase); (7) - (8) Need for further training for employees in same occupation as oneself (dummy coded one if increase); Randomized experimental treatment group "Information": respondents informed about automatability of occupation. Using the full variation of the 5-point Likert scale does not change the interpretation of the results; Baseline mean: Mean of the variable in the control group or baseline group. See Table 1 for included covariates. Outcomes in this table are pre-registered as secondary outcomes. Data source: ifo Education Survey 2022. Robust standard errors in parentheses. Significance levels: * p < 0.1, ** p < 0.05, *** p < 0.01.

Appendix Table A6: Barriers to training participation (I)

	Financial constraints		Time constraints		Employer constraints		Offered by FEA	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Information	0.014	0.035	0.003	0.020	-0.013	-0.018	-0.001	0.004
	(0.018)	(0.025)	(0.017)	(0.024)	(0.018)	(0.025)	(0.018)	(0.025)
Low Automation		0.003		0.008		-0.039		0.008
		(0.025)		(0.024)		(0.025)		(0.025)
Information x Low Automation		-0.041		-0.035		0.011		-0.010
		(0.035)		(0.034)		(0.035)		(0.035)
Covariates	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Baseline Mean	0.454	0.457	0.352	0.345	0.450	0.473	0.395	0.390
Observations	3009	3009	3010	3010	3011	3011	3011	3011
R-squared	0.068	0.068	0.046	0.046	0.074	0.075	0.026	0.026
Information Effect for High Automation		-0.006		-0.015		-0.007		-0.006
		(0.025)		(0.024)		(0.025)		(0.025)

Notes: OLS regressions. Dependent variables (dummy = 1 if person fully or somewhat agrees): (1) - (2) I cannot financially afford to attend further training; (3) - (4) I do not have time for further training (e.g., because of caring for relatives, childcare, etc.); (5) - (6) My employer does not offer me the opportunity for professional development; (7) - (8) I do not wish to participate in any further vocational training funded by the Federal Employment Agency (FEA); Randomized experimental treatment group "Information": respondents informed about automatability of occupation. Baseline mean: Mean of the variable in the control group or baseline group. See Table 1 for included covariates. Outcomes in this table are pre-registered as secondary outcomes. Data source: ifo Education Survey 2022. Robust standard errors in parentheses. Significance levels: * p < 0.1, ** p < 0.05, *** p < 0.01.

	Confident about professional future		Great necessity to partie	cipate in further training	Unsure about returns to training	
	(1)	(2)	(3)	(4)	(5)	(6)
Information	-0.007	-0.020	-0.005	0.026	0.010	0.032
	(0.017)	(0.024)	(0.018)	(0.025)	(0.018)	(0.026)
Low Automation		0.014		0.069***		-0.000
		(0.024)		(0.025)		(0.026)
Information x Low Automation		0.027		-0.063*		-0.044
		(0.034)		(0.035)		(0.036)
Covariates	Yes	Yes	Yes	Yes	Yes	Yes
Baseline Mean	0.665	0.658	0.475	0.435	0.476	0.481
Observations	3011	3011	3010	3010	3011	3011
R-squared	0.052	0.053	0.069	0.071	0.019	0.020
Information Effect for Low Automation		0.007		-0.037		-0.012
		(0.024)		(0.026)		(0.026)

Appendix Table A7: Barriers to training participation (II)

Notes: OLS regressions. Dependent variables (dummy = 1 if person fully or somewhat agrees): (1) - (2) I am unsure whether further training will pay off for me; (3) - (4) I see a great need to participate in further training; (5) - (6) I am well equipped for my future; Randomized experimental treatment group "Information": respondents informed about automatability of occupation. Baseline mean: Mean of the variable in the control group or baseline group. See Table 1 for included covariates. Outcomes in this table are pre-registered as secondary outcomes. Data source: ifo Education Survey 2022. Robust standard errors in parentheses. Significance levels: * p < 0.1, ** p < 0.05, *** p < 0.01.

Appendix Table A8: Information effect on information acquisition

	Information acquisition	Information acquisition		
	(1)	(2)		
Information	0.002	0.029		
	(0.017)	(0.024)		
Low Automation		0.059**		
		(0.025)		
Information x Low Automation		-0.056		
		(0.035)		
Covariates	Yes	Yes		
Baseline Mean	0.386	0.350		
Observations	3010	3010		
R-squared	0.066	0.068		
Information Effect for Low Automation		-0.027		
		(0.025)		

Notes: OLS regressions. Dependent variables: (1) - (2) Information acquisition; Randomized experimental treatment group "Information": respondents informed about automatability of occupation. Baseline mean: Mean of the variable in the control group or baseline group. See Table 1 for included covariates. Outcomes in this table are pre-registered as secondary outcomes. Data source: ifo Education Survey 2022. Robust standard errors in parentheses. Significance levels: * p < 0.1, ** p < 0.05, *** p < 0.01.